Author’s response to reviews

Title: Body size perceptions and preferences favor overweight in adult Saharawi refugees

Authors:

Desire Naigaga (Desire.Naigaga@hioa.no)
David Jahanlu (djahanlu@me.com)
Hanne Marit Claudius (hmclaudius@gmail.com)
Anne Karine Gjerlaug (annekarine.gjerlaug@hioa.no)
Ingrid Barikmo (ingrid.barikmo@outlook.com)
Sigrun Henjum (sigrun.henjum@hioa.no)

Version: 3 Date: 17 Jan 2018

Author’s response to reviews:

Reviewer #1: Dear authors,

You have met all recommendations. I have only one alteration that should be considered. Table 3 is confusing because you have body discrepancy, self-perceived body size and desired body size, and it is not possible to understand what the columns and rows refer to.

Reply: We agree and have changed table 3 according to your suggestion.

Please also double check the quote added to the introduction - is this meant to be a direct quote?

Reply: Thank you for making us aware of this we have removed the quote.